Author's response to reviews

Title: Clinicopathological features of KRAS, NRAS, BRAF and PIK3CA mutations in Japanese patients with metastatic colorectal cancer

Authors:

Akihito Kawazoe (akawazoeakihito@gmail.com)
Kohei Shitara (kshitara@east.ncc.go.jp)
Shota Fukuoka (shfukuok@east.ncc.go.jp)
Yasutoshi Kuboki (ykuboki@east.ncc.go.jp)
Hideaki Bando (hbando@east.ncc.go.jp)
Wataru Okamoto (wokamoto@east.ncc.go.jp)
Takashi Kojima (takojima@east.ncc.go.jp)
Nozomu Fuse (nofuse@east.ncc.go.jp)
Takeharu Yamanaka (yamanaka@yokohama-cu.ac.jp)
Toshihiko Doi (tdoi@east.ncc.go.jp)
Atsushi Ohtsu (aohtsu@east.ncc.go.jp)
Takayuki Yoshino (tyoshino@east.ncc.go.jp)

Version: 2 Date: 30 September 2014

Author's response to reviews:

Dear Editor-in-Chief,

Thank you for your suggestion of revisions in my manuscript.

According to your suggestion, we added the email addresses of all authors in a title page at the front of my manuscript. In addition, we added a section of Acknowledgements and the sentence of the approval of an appropriate ethics committee.

Sincerely,

Takayuki Yoshino, M.D.

Department of Gastroenterology and Gastrointestinal Oncology, National Cancer Center Hospital East, 6-5-1 Kashiwanoha, Kashiwa, Chiba 277-8577, Japan.